Author’s response to reviews

Title: Evaluation of Risk Factors for Perforated Peptic Ulcer

Authors:

Kazuki Yamamoto (kazuyama@luke.ac.jp)
Osamu Takahashi (otakahas@luke.ac.jp)
Hiroko Arioka (arihiro@luke.ac.jp)
Daiki Kobayashi (daikoba@luke.ac.jp)

Version: 3 Date: 05 Feb 2018

Author’s response to reviews:

Dear Editor

Subject: Submission of revised paper “Evaluation of Risk Factors for Perforated Peptic Ulcer”, ID: BMGE-D-17-00333

We thank referees for careful reading our manuscript and for giving useful comments. We feel the comments will significantly improve our paper. In response to the Referees' comments, we have revised the manuscript.

Please take a look our revised manuscript.

Thank you very much

Sincerely,

Kazuki Yamamoto, MD
St. Luke’s International Hospital, 9-1 Akashi-cho, Chuo-ku, Tokyo 104-8560, Japan
Telephone number: +81-3-5550-7120